Author's response to reviews

Title: Retention of Healthcare Professionals in Urban India Using Health Information Technology - A Path Analysis Study

Authors:

Indrajit Bhattacharya (indrajitbhat@gmail.com)
Anandhi Ramachandran (anandhirama@gmail.com)

Version: 2 Date: 17 March 2014

Author's response to reviews:

Editor,
Journal of Human Resource of Health

Human resource in health has been a challenging area in the developing countries, and is extremely acute in India in comparison to other peer countries. The attrition and the retention strategies have always remained a long standing issue to resolve. This study is relevant as the ‘healthcare-IT ‘ has been taken as one of the measurement methodologies for retention strategy. After statistical analysis, a unique ‘Path analysis’ study has been carried out to develop a model for retention strategy.

This study would certainly provide a retention strategy perspective for planners of human resource knowledge workers in healthcare in India.

Revised version is as per instructions of manuscript submission.